Author's response to reviews

Title: Control of asthma triggers in indoor air with air cleaners: a modeling analysis

Authors:

  Theodore A Myatt (tmyatt@eheinc.com)
  Taeko Minigishi (tminigishi@eheinc.com)
  Joseph G Allen (jallen@eheinc.com)
  David L MacIntosh (dmacintosh@eheinc.com)

Version: 5  Date: 31 July 2008

Author's response to reviews:

Dear Dr. Ozonoff:

We appreciate the comments of the one of the reviewers. We had no issues with any of the suggestions and therefore have provided a revised manuscript with the changes incorporated. We think it is important to note, that contrary to the reviewer’s suggestion, the interests of the funder did not motivate the statement in the abstract regarding the allergen avoidance strategies by organizations such as ALA and NIH. While the funder has seen a draft of this manuscript, we never received any comments or suggestions regarding the content of the analysis or language in the draft.

Thank you again for your consideration, Ted Myatt